Author's response to reviews

Title: Smoking differences between employees in faculties of the University of Tartu, Estonia, and changes during the country's transition

Authors:

Simo Näyhä (simo.nayha@oulu.fi)
Jana Kivastik (jana.kivastik@ut.ee)
Peet-Henn Kingisepp (peet-henn.kingisepp@ut.ee)
Rauno Heikkinen (rauno.heikkinen@oulu.fi)

Version: 5 Date: 7 March 2011

Author's response to reviews:

Editorial Office March 7, 2011
BMC Public Health

Your ref.: MS: 9882049639085184

Näyhä S, Kivastik J, Kingisepp PH, Heikkinen R. Smoking differences between employees in faculties of the University of Tartu, Estonia, and changes during the country's transition

Dear Sir/Madam

Here we are submitting the final version of the above manuscript which has been checked as requested by the BMC Series Editorial Production Team (email Mar 1, 2011). Some typing errors have been corrected, and the figures, tables and additional files have been checked for errors (there were some).

Yours sincerely,

Simo Näyhä
Professor, M.D., Ph.D.
Department of Health Sciences
University of Oulu
P.O. Box 5000
90014 University of Oulu
Finland
Phone +358-8-5375 671
Mobile +358-40-5646 159
Fax +358-8-5375 661